Author’s response to reviews

Title: Prevalence and characteristics of COPD among pneumoconiosis patients at an occupational disease prevention institute: A cross-sectional study

Authors:

yating peng (pengyating2011@126.com)
xin li (40707074@qq.com)
shan cai (caishan70@sina.com)
yan chen (chenyan99727@163.com)
weirong dai (40404814@qq.com)
wenfeng liu (314095618@qq.com)
zijing zhou (zhouzijing1114@126.com)
ping chen (pingchen00731@126.com)

Version: 1 Date: 26 Oct 2017

Author’s response to reviews:

Dear reviewer:

Thank you for your suggestion. We already ask American Journal Experts for help with language editing of our manuscript. We confirm that all author details on the revised version are correct, all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data. We also rearranged the Declarations section with sub-sections according to journal's Submission Guidelines. Thank you again!

Ping Chen